Author's response to reviews

Title: Handling Health Promotion in Hospitals in diagnosis-related groups (DRGs): evaluation of systematic codes for health promotion activities in a multicentre project

Authors:

Hanne Tonnesen (ht02@bbh.hosp.dk)
Mette E Christensen (metschen@yahoo.dk)
Oliver Groene (ogr@es.euro.who.int)
Ann O'Riordan (info@ihph.ie)
Fabrizio Simonelli (f.simonelli@meyer.it)
Lagle Suurorg (lage.suurorg@lastehaigla.ee)
Denise Morris (dmorris@uclan.ac.uk)
Peder Vibe (Peder.Vibe@skane.se)
Poul Erik Hansen (PEH@sst.dk)
Susan Himel (shimel@thc.on.ca)

Version: 6 Date: 25 July 2007

Author's response to reviews:

The title of co-author Fabrizio Simonelli has been changed from MD to MSc
The title MES of co-author Susan Himel has been added